\documentclass{article}
\usepackage[affil-it]{authblk}
\usepackage{graphicx}
\usepackage[space]{grffile}
\usepackage{latexsym}
\usepackage{textcomp}
\usepackage{longtable}
\usepackage{tabulary}
\usepackage{booktabs,array,multirow}
\usepackage{amsfonts,amsmath,amssymb}
\providecommand\citet{\cite}
\providecommand\citep{\cite}
\providecommand\citealt{\cite}
\usepackage{url}
\usepackage{hyperref}
\hypersetup{colorlinks=false,pdfborder={0 0 0}}
\usepackage{etoolbox}
\makeatletter
\patchcmd\@combinedblfloats{\box\@outputbox}{\unvbox\@outputbox}{}{%
  \errmessage{\noexpand\@combinedblfloats could not be patched}%
}%
\makeatother
% You can conditionalize code for latexml or normal latex using this.
\newif\iflatexml\latexmlfalse
\AtBeginDocument{\DeclareGraphicsExtensions{.pdf,.PDF,.eps,.EPS,.png,.PNG,.tif,.TIF,.jpg,.JPG,.jpeg,.JPEG}}

\usepackage[utf8]{inputenc}
\usepackage[english]{babel}

\newcommand{\truncateit}[1]{\truncate{0.8\textwidth}{#1}}
\newcommand{\scititle}[1]{\title[\truncateit{#1}]{#1}}



\begin{document}

\title{Detecting artificial satellites around exoplanets}


\author{Matteo Cantiello}
\affil{Affiliation not available}

  
\author{Eli Bressert}
\affil{Affiliation not available}
  



\date{\today}

\maketitle

\selectlanguage{english}
\begin{abstract}
The search for extrasolar planets (EXOPLANETS) is one of the most rapidly expanding fields in astrophysics. Thanks to recent space-based efforts the number of detected planets in extrasolar stellar systems has increased dramatically, and more than 900 objects have been identified as of today, see e.g. \url{http://exoplanet.eu/catalog/}. With candidates identified in the so-called habitable zone (HZ)\footnote{The theoretical band around a star where a planet could orbit and host liquid water} the question of extraterrestrial life becomes very actual. Here we discuss ideas that could lead to infer the presence (or the past existence) of an extraterrestrial civilization by the detection of artifacts in orbit around the host planet. %
\end{abstract}%



\begin{quote}
This is an attempt to write an {\bf Authorea Open Science} article. The author of the seed abstract (seed author) welcomes any collaborator willing to give comments / input on the topic discussed. For the time being people who want to be added as collaborators need to send an email request to the seed author (matteo at kitp.ucsb.edu). As the effort progress, the different contributions are logged in and stored in the Authorea version system, so that in the final version of the paper the contributions from the various authors can be viewed in detail and eventually a ranked author list produced. The final article will be published in Authorea (and possibly a high impact factor journal) and posted on AstroPh. 
\end{quote}

\section{Topics} 
\begin{enumerate}
\item Geometry
\item Solar panels (Need somebody expert in the properties of {\bf materials} usually adopted to build solar panels)
\item Detection in polarized light (Need an {\bf expert on polarization})
\item Transit signature (Need an {\bf expert on transit light curves})
\item How to distinguish moons from artificial satellites? (adopt combination of polarization and transit signature)
\item Spectra?
\item Test: signal from International Space Station as seen from 10 pc. How that scales with size of artificial satellite?
\item Literature. Need somebody from the field (Exoplanets) that can point to the {\bf relevant literature }
\end{enumerate}

\section{Some literature}
\cite{2011ApJ...741...51R}

\cite{2011ApJ...741...51R}

\cite{2011AAS...21731805M}

\cite{2008ApJ...685L.153L}

\cite{arxiv.org/pdf/1303.1100.pdf}

\cite{arxiv.org/abs/1304.2925}

\selectlanguage{english}
\FloatBarrier
\end{CJK}\end{document}

